Author’s response to reviews

Title: Symptoms of depression are associated with physical inactivity but not modified by gender or the presence of a cardiovascular disease; a cross-sectional study.

Authors:

rj achttien (retze.achttien@gmail.com)
Jan van Lieshout (Jan.vanLieshout@radboudumc.nl)
Michel Wensing (Michel.Wensing@radboudumc.nl)
Ria Nijhuis-van der Sanden (Ria.Nijhuis-vanderSanden@radboudumc.nl)
Bart Staal (Bart.Staal@radboudumc.nl)

Version: 2 Date: 18 Mar 2019

Author’s response to reviews:

Dear editor,

We are grateful that you have assessed the manuscript as potentially acceptable for publication. We have carried out the proposed revisions. We specified the individual contributions of all authors to the manuscript in the Authors’ Contributions section. We removed the additional files and uploaded the manuscript as a single, final, clean version (including the tables).

On behalf of the authors,

Regards,

Retze Achttien